Author’s response to reviews

Title: Prevalence and risk factors for Plasmodium falciparum malaria in pregnant women attending antenatal clinic in Bobo-Dioulasso (Burkina Faso).

Authors:

Mamoudou Cisse (cisse_m@yahoo.fr)
Ibrahim Sangare (babaibrasangare@yahoo.fr)
Guekoun Lougue (glouque5@yahoo.fr)
Sanata Bamba (hsanata@yahoo.fr)
Dramane Bayane (bayanedramane@yahoo.fr)
Robert Tinga Guiguemde (rguiguemde@yahoo.fr)

Version: 2
Date: 30 April 2014

Author’s response to reviews: see over
To The Editorial Board of *BMC Infectious Diseases*

April 30, 2014

Dear Editor,

We hereby acknowledge having included line numbers in our manuscript as recommended. The page numbers have been previously included in the manuscript.

We look forward for hearing from you.

Sincerely,

Dr. Mamoudou Cisse  
MD, MPhil, PhD student  
Malaria and neglected tropical diseases research unit  
Centre MURAZ research institute  
Bobo-Dioulasso, Burkina Faso  
cisse_m@yahoo.fr